Title: Overcoming recruitment barriers revealed high readiness to participate and low dropout rate among people with schizophrenia in a randomized controlled trial testing the effect of a Guided Self-Determination intervention.

Authors:

Rikke Jørgensen (rijo@rm.dk)
Povl Munk-Jørgensen (povlmunk@rm.dk)
Paul H Lysaker (plysaker@iupui.edu)
Kelly D Buck (Kelly.Buck@va.gov)
Lars Hansson (Lars.Hansson@med.lu.se)
Vibeke Zoffmann (vzof@steno.dk)

Version: 2 Date: 17 December 2013

Author's response to reviews:

Cover letter for the referees:
Thank you very much for your valuable comments. Specifically, the comments from referee 2 regarding the focus and the structure of the manuscript started a reflection process, which resulted in a decision to reconsider the entire manuscript.

Agreeing with the referee’s comments and concerns, the manuscript has been rewritten entirely with each referees’ comments and concerns in mind. No point-by-point response to the referees’ comments and concerns has been made as we have taken them all into consideration in our rewriting.

With my best regards on behalf of all authors,

Rikke Jørgensen

PhD, PostDoc

Unit for Psychiatric Research

Aalborg Psychiatric Hospital

Denmark